Author's response to reviews

Title: Estrogen regulation of TRPM8 expression in breast cancer cells

Authors:

Dechen Chodon (dechenibms@rediffmail.com)
Arnaud Guilbert (arnaud.guilbert@gmx.com)
Isabelle Dhennin-Duthillea (isabelle.dhennin@u-picardie.fr)
Mathieu Gautier (mathieu.gautier@u-picardie.fr)
Marie-Sophie Telliez (marie-sophie.telliez@u-picardie.fr)
Henri Sevestre (sevestre.henri@chu-amiens.fr)
Halima Ouadid-Ahidouch (ha-sciences@u-picardie.fr)

Version: 3 Date: 26 March 2010

Author's response to reviews: see over
Dear Dr Diana Marshall

We are returning the revised version of our manuscript entitled “ESTROGEN REGULATION OF TRPM8 EXPRESSION IN BREAST CANCER CELLS” prepared according to the BMC cancer.

The minor comments of the referees 3, 4 and 2 were taken into consideration and the text was revised accordingly.

Sincerely,

Halima Ouadid-Ahidouch, Pr